

Citation: Michel LJ, Rospleszcz S, Reisert M, Rau A, Nattenmueller J, Rathmann W, et al. (2024) Deep learning to estimate impaired glucose metabolism from Magnetic Resonance Imaging of the liver: An opportunistic population screening approach. PLOS Digit Health 3(1): e0000429. https://doi.org/10.1371/journal.pdig.0000429

Editor: Martin G. Frasch, University of Washington, UNITED STATES

Received: July 15, 2023

Accepted: December 7, 2023

Published: January 16, 2024

Copyright: © 2024 Michel et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The informed consent given by KORA study participants does not cover data posting in public databases. However, data are available upon request by means of a project agreement. Requests should be sent to kora.passt@helmholtz-muenchen.de and are subject to approval by the KORA Board.

Funding: The KORA study was initiated and financed by the Helmholtz Zentrum München – German Research Center for Environmental Health,

RESEARCH ARTICLE

Deep learning to estimate impaired glucose metabolism from Magnetic Resonance Imaging of the liver: An opportunistic population screening approach

Lea J. Michel^{1*}, Susanne Rospleszcz^{2,3,4}, Marco Reisert⁵, Alexander Rau¹, Johanna Nattenmueller¹, Wolfgang Rathmann⁶, Christopher. L. Schlett¹, Annette Peters^{2,3,4,7}, Fabian Bamberg¹, Jakob Weiss¹

 Department of Diagnostic and Interventional Radiology, University Hospital Freiburg, Freiburg, Germany,
Department of Epidemiology, Institute for Medical Information Processing, Biometry, and Epidemiology, Ludwig-Maximilians-University Munich, Munich, Germany, 3 Institute of Epidemiology, Helmholtz Zentrum München, German Research Center for Environmental Health, Neuherberg, Germany, 4 German Center for Cardiovascular Research (DZHK), partner site Munich Heart Alliance, Germany, 5 Medical Physics, Department of Radiology, Medical Center—University of Freiburg, Freiburg, Germany, 6 Institute for Biometrics and Epidemiology, German Diabetes Center, Leibniz Center for Diabetes Research, Heinrich Heine University Düsseldorf, Düsseldorf, Germany; German Center for Diabetes Research, München-Neuherberg, Germany, 7 German Center for Diabetes Research (DZD), partner site Neuherberg, Neuherberg, Germany

* lea.michel@uniklinik-freiburg.de

Abstract

Aim

Diabetes is a global health challenge, and many individuals are undiagnosed and not aware of their increased risk of morbidity/mortality although dedicated tests are available, which indicates the need for novel population-wide screening approaches. Here, we developed a deep learning pipeline for opportunistic screening of impaired glucose metabolism using routine magnetic resonance imaging (MRI) of the liver and tested its prognostic value in a general population setting.

Methods

In this retrospective study a fully automatic deep learning pipeline was developed to quantify liver shape features on routine MR imaging using data from a prospective population study. Subsequently, the association between liver shape features and impaired glucose metabolism was investigated in individuals with prediabetes, type 2 diabetes and healthy controls without prior cardiovascular diseases. K-medoids clustering (3 clusters) with a dissimilarity matrix based on Euclidean distance and ordinal regression was used to assess the association between liver shape features and glycaemic status.

Results

The deep learning pipeline showed a high performance for liver shape analysis with a mean Dice score of 97.0±0.01. Out of 339 included individuals (mean age 56.3±9.1 years; males

which is funded by the German Federal Ministry of Education and Research (BMBF) and by the State of Bavaria. Data collection in the KORA study is done in cooperation with the University Hospital of Augsburg. This study was funded by German Research Foundation (Bonn, Germany) grant BA 4233/4-1. For this sub-study the authors received no specific funding. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

58.1%), 79 (23.3%) and 46 (13.6%) were classified as having prediabetes and type 2 diabetes, respectively. Individuals in the high risk cluster using all liver shape features (n = 14) had a 2.4 fold increased risk of impaired glucose metabolism after adjustment for cardiometabolic risk factors (age, sex, BMI, total cholesterol, alcohol consumption, hypertension, smoking and hepatic steatosis; OR 2.44 [95% Cl 1.12–5.38]; p = 0.03). Based on individual shape features, the strongest association was found between liver volume and impaired glucose metabolism after adjustment for the same risk factors (OR 1.97 [1.38–2.85]; p < 0.001).

Conclusions

Deep learning can estimate impaired glucose metabolism on routine liver MRI independent of cardiometabolic risk factors and hepatic steatosis.

Author summary

Type 2 diabetes presents a significant global health challenge, affecting an estimated 536.6 million individuals, comprising 10.5% of the adult population worldwide. Alarmingly, nearly half of those affected remain undiagnosed, unaware of their elevated risk of morbidity and mortality. Although cheap and accurate tests exist, the high number of undiagnosed individuals demonstrates the need for novel approaches. Especially in medical imaging, we have large amounts of data acquired every day, yet most of the information embedded in the scans remains unused. With deep learning, it is now possible to automatically extract this information in an opportunistic fashion. For this study a fully automatic deep learning framework was developed to investigate the association between radiomic shape features of the liver and impaired glucose metabolism in a sample from a population-based cohort of individuals with prediabetes, type 2 diabetes and healthy controls without prior cardiovascular diseases. Our results demonstrate that the proposed deep learning framework allows for identifying individuals at risk for impaired glucose metabolism on routinely acquired MRI of the liver. These findings indicate the potential for opportunistic screening to identify individuals at risk of type 2 diabetes to improve screening and early diagnosis, to reduce long-term complications and economic expenses.

Introduction

Type 2 diabetes is defined as a relative insulin deficiency resulting from defects in the pancreatic betta cells and insulin resistance in target organs [1]. During the past decades, the prevalence has dramatically increased due to rapid urbanisation and a more sedentary lifestyle [2,3,4]. Currently, the global prevalence of diabetes counts 536.6 million people or 10.5% of the adult population and is considered to further increase with an estimate of 783.2 million diseased individuals by 2045 [4]. This is of great importance, as the management of type 2 diabetes and its complications is placing a substantial economic burden on the healthcare system [5]. For example, Jacobs et al. revealed, that in Germany, 10% of the total salutary health insurance expenses, in total 16.1 billion euros, were attributed to the medical care of type 2 diabetes in 2017 [6]. In the same year, the total expenditure on diabetes treatment in the United States was 327 billion dollars [7,8]. In addition, there is a high number of individuals with undiagnosed diabetes, who are not aware of their increased risk of morbidity and mortality because disease onset is typically subclinical and asymptomatic [9]. Although dedicated tests for screening and early diagnosis are available, such as glycated haemoglobin (HbA1c) or oral glucose tolerance testing, correct interpretation may be confounded with the risk of reduced accuracy [10,11,12,13]. Further, non-invasive risk scores and biomarkers such as the QDScore or DPoRT are valuable tools to identify individuals at high risk of type 2 diabetes in the general population but have been shown to perform sub-optimally at an individual level [14,15,16,17]. This demonstrates the need for novel approaches to improve screening and early diagnosis to reduce long-term complications and economic expenses.

Recent developments in deep learning–a form of artificial intelligence, have opened a new window for high-throughput quantitative analysis. This is particularly true for medical imaging, where large amounts of data are acquired every day, yet most of the information embedded in the scans remains unused due to time and equipment constraints. With deep learning, it is now possible to automatically extract this information in an opportunistic fashion from scans acquired in daily care with the potential to improve clinical decision-making and patient management.

Here, we developed a fully automatic deep learning framework to quantify radiomic shape features of the liver on routine MRI to estimate states of impaired glucose metabolism in a sample from a population-based cohort. We hypothesize, that MRI of the liver, a key organ in glucose metabolism and homeostasis, contains features indicative of impaired glucose metabolism independent of traditional cardiometabolic risk factors and hepatic steatosis and allows for opportunistic screening of individuals at risk of impaired glucose metabolism.

Material and methods

Study population

All analyses were performed using data from the Cooperative Health Research in the Region of Augsburg (KORA) MRI study, a cross-sectional MRI study nested in the prospective epidemiological cohort of the KORA main study [18,19]. The study population of the KORA-MRI study was recruited between June 2013 and September 2014 and consisted of 400 participants who underwent a multiparametric whole-body MRI study protocol as previously described in detail [19]. Subjects were eligible if they met the following inclusion criteria: consent to undergo whole-body MRI, no prior cardiovascular disease, and a standardized assessment of glucose metabolism.

All participants gave written informed consent, and the study protocol was approved by the ethics committee of the Bavarian Chamber of Physicians and the institutional review board of the Ludwig-Maximilians-University Munich and complied with Helsinki's declaration of human research.

MRI protocol

Whole-body MRIs were acquired on a 3T MRI scanner (Magnetom Skyra, Siemens Healthcare, Erlangen, Germany) in supine position using an 18-channel body surface coil and a table mounted spine matrix coil. The complete MRI protocol has been described previously [19]. For the current study, only the two-point T1-weighted isotropic Volumetric interpolated breath-hold examination (VIBE)—Dixon gradient-echo sequence was used, which was acquired with the following acquisition parameters: slice thickness: 1.7 mm, in-plane resolution 1.7x1.7 mm², field of view: 488x716 mm using a 256x256 matrix, repetition time: 4.06 ms and echo time: 1.26x2.49 ms; flip angle 9° [19]. For all further analyses, water images were used.

Deep learning framework for liver shape analysis

We developed a fully automatic framework to quantify radiomic shape features of the liver on T1-weighted MRI. The framework comprised the following steps: 1) deep learning model for 3D liver segmentation. The only input to the model was a T1-weighted isotropic VIBE-Dixon gradient-echo sequence using the water contrast reconstruction and the output was a 3D segmentation mask of the liver. 2) Quantification of radiomic shape features based on the 3D liver segmentation mask of the deep learning model. Radiomic features were extracted using PyRadimoics (https://pyradiomics.readthedocs.io/). The entire framework was implemented in NORA (www.nora-imaging.com), an open-source medical imaging platform.

Deep learning model for liver segmentation

For automated liver segmentation, we employed a hierarchical, multiscale 3D convolutional neural network, which uses nested patches of fixed matrix size (https://bitbucket.org/reisert/ patchwork/) [20]. In each scale, a UNet-like architecture is used, where the matrix size of the UNet is always of size 32^3 voxels for all scales. We used a scale pyramid of depth four, where the patch size of the coarsest scale is a cube with side length of 80% of the full image. The physical size of the finest patch is selected such that a resolution 1.5, 1.5, 3 mm is achieved at the final output. As input, we stacked the derived water images of the T1-weighted Dixon sequence. The architecture of the basis UNet is close to the method described by Cicek et al., 2016 [21], with feature dimensions (8,16,16,32,64) and max-pooling in the encoding layers and transposed convolutions in the decoding layers. Each UNet has n+8 output channels, where the first n is the number of labels and are used for intermediate loss computations. The logits of the total n+8 outputs are just forwarded as input to the next scale. The network is trained with the Adam optimizer with a learning rate 0.001. All labels are trained with ordinary binary crossentropy per channel. The prediction of a full volume was performed on a 16-core machine without GPU support and takes less than 5 minutes. All data management and processing were performed using NORA (www.nora-imaging.com).

For model development, a random subset of 124 participants was used. Segmentation labels of the liver were manually generated by a trained radiologist and proofed read/adapted if necessary, by an additional board-certified radiologist. For model training, a random subset of 80 participants was used. Independent testing was performed in the remaining, held-out 44 participants not seen during any part of training. Model performance was evaluated using the DICE index, the average surface distance and Pearsons's correlation coefficient.

Subsequently, the model was applied to all participants of the KORA-MRI study. For additional quality control, the 3D liver segmentation masks of all participants were visually validated by a trained radiologist and excluded from further analyses, if substantial errors were detected.

Radiomic feature extraction

Based on the 3D liver segmentation mask we extracted 14 radiomic shape features (e.g. volume, sphericity) using the open-source software PyRadiomics (version 3.0). Detailed descriptions of the individual features are provided on the PyRadiomics homepage (<u>https://</u> pyradiomics.readthedocs.io/en/latest/features.html).

Assessment of impaired glucose metabolism and other clinical covariates

Baseline demographics and cardiometabolic risk factors of the study participants were evaluated in detailed standardized interviews and standardized medical examination as described previously in detail [18]. Variables of interest included age in years, sex, body mass index (BMI; kg/m²), weight (kg), height (cm), waist circumference (cm), hypertension (mmHg, defined as systolic blood pressure of at least 140 mmHg, diastolic blood pressure of at least 90 mmHg), antihypertensive medication, glycated haemoglobin (HbA1c %), total cholesterol (mg/dl), low density lipoprotein-C (mg/dl), triglycerides, (mg/dl), hepatic steatosis (defined as MRI-derived hepatic fat fraction >5% [19]), alcohol consumption (g/day) and smoking status (never, former, current).

Outcome

The primary outcome of this study was impaired glucose metabolism defined according to the 1998 World Health Organization criteria [22] and assessed via an oral glucose tolerance test. Per study protocol, individuals with a 2-h serum glucose concentration >200 mg/dl and/or a fasting glucose level >125 mg/dl were classified as having type 2 diabetes, participants with a 2-h serum glucose concentration between 140 and 200 mg/dL; and/or an impaired fasting glucose concentration between 110 and 125 mg/dL as having prediabetes [19]. Participants with previously established and physician-validated type 2 diabetes did not undergo oral glucose test but were immediately classified as having type 2 diabetes.

Statistical analysis

All statistical analysis was conducted using R version 4.1.2 and Microsoft Excel version 16.66.1. Continuous variables are described as mean±SD (standard deviation) or median and interquartile range (IQR) as appropriate. Categorical variables are provided as frequencies and percentages, respectively. Differences between demographics and cardiometabolic risk factors were tested using Student's t-test, Wilcoxon rank-sum text or x2-test as appropriate.

To investigate the association between radiomic shape features of the liver and impaired glucose metabolism, a two-step approach was conducted. 1) Clustering approach: to test the combined association of all 14 extracted shape features of the liver and impaired glucose metabolism, we used k-medoids clustering (3 clusters) with a dissimilarity matrix based on Euclidean distance to derive unbiased data-driven clusters from the shape features only. Clusters were used as categorical exposures (reference: first cluster) in an ordinal logistic regression model with proportional odds assumption and outcomes prediabetes and diabetes. We estimated univariable and multiple adjusted Odds Ratios (OR), with stepwise adjustment for baseline demographics (age, sex, BMI) and cardiometabolic risk factors (total cholesterol, alcohol consumption, hypertension, smoking status, hepatic steatosis). 2) Individual approach: to investigate the association between individual shape features and impaired glucose metabolism, they were used as continuous exposures in ordinal logistic regression with adjustments as outlined above. Odds Ratios are reported per 1 SD change in shape feature. Liver volume was defined as the key feature of interest, as it is often estimated in clinical routine and easier to grasp than e.g. elongation or sphericity. To limit the analyses to features with the highest effect size and avoid redundancy with liver volume, only features with a univariable OR \geq 1.5 and a correlation ≤ 0.5 with liver volume were considered. All p-values are two-sided and are considered to denote statistical significance if < 0.05.

Results

Study population

An overview of the study design is presented in Fig 1. Of the 400 participants included in the KORA-MRI study, 46 participants were excluded due to missing or corrupted imaging data. Additional 15 participants were excluded due to missing covariates resulting in a final study cohort of 339 individuals (Fig 2) with a mean age of 56.3 ± 9.1 years. 58.1% were male and 23.3% (79/339) vs. 13.6% (46/339) were classified as having prediabetes and type 2 diabetes, respectively. Participants with type 2 diabetes were significantly older, more likely men and presented more often with prevalent hypertension and hepatic steatosis compared to normoglycemic controls (all p<0.001). Further detailed demographics are provided in Table 1.



Fig 1. Overview of the study design. A) The deep learning framework for fully automated extraction of liver shape features was developed in participant of the KORA MRI study. B) The performance of the framework was tested in an independent random subset of 44 participants not seen during any part of model development. In addition, the predictive value of liver shape features for impaired glucose metabolism was investigated in the entire study cohort. C) Representative case of the independent testing dataset with manual (red) and automatically (green) generated liver segmentations. KORA = Cooperative Health Research in the Region of Augsburg; CVD = Cardiovascular Disease; MRI = Magnetic Resonance Imaging.

https://doi.org/10.1371/journal.pdig.0000429.g001



Fig 2. Participant flowchart of the KORA-MRI study. Other reason: 1 withdrew consent; 4 missing values of liver shape; 4 missing values of hepatic steatosis; 6 missing covariates.

https://doi.org/10.1371/journal.pdig.0000429.g002

Table 1.	Baseline dem	ographics and	cardiovascul	ar risk factors.

Variable	Entire cohort	Normoglycemia	Prediabetes	Diabetes	p-value
	N = 339	N = 214 (63.1%)	N = 79 (23.3%)	N = 46 (13.6%)	
Mean age, years±SD	56.3 ± 9.1	54.1 ± 8.8	58.6 ± 8.6	62.5 ± 7.6	<0.001
Male sex	197 (58.1%)	108 (50.5%)	53 (67.1%)	36 (78.3%)	<0.001
BMI, kg/m ²	28.0 ± 4.7	26.7 ± 4.3	30.4 ± 4.4	29.6 ± 4.9	<0.001
Weight, kg	82.9 ± 16.3	78.8 ± 15.8	91.1 ± 13.9	87.5 ± 16.5	<0.001
Height, cm	171.9 ± 9.7	171.4 ± 10.4	173.2 ± 8.8	171.7 ± 7.8	0.391
Waist circumference, cm	98.3 ± 13.8	93.8 ± 12.8	106.0 ± 10.8	106.2 ± 13.5	<0.001
Hypertension	112 (33.0%)	45 (21.0%)	36 (45.6%)	31 (67.4%)	<0.001
Antihypertensive medication	82 (24.2%)	36 (16.8%)	25 (31.6%)	21 (45.7%)	<0.001
HbA1c, %	5.6 ± 0.8	5.3 ± 0.3	5.6 ± 0.3	6.8 ± 1.4	<0.001
Total Cholesterol, mg/dL	218.8 ± 37.3	216.4 ± 36.1	227.2 ± 32.1	215.2 ± 48.0	0.068
LDL-C, mg/dL	140.6 ± 33.9	138.9 ± 32.0	148.0 ± 31.8	135.6 ± 43.3	0.072
Triglycerides, mg/dL	131.0 ± 85.7	105.3 ± 60.7	158.7 ± 85.6	202.8 ± 123.5	<0.001
Hepatic Steatosis	188 (55.5%)	85 (39.7%)	64 (81.0%)	39 (84.8%)	<0.001
Alcohol consumption, g/day [median (CI)]	8.6 [0.5, 26.8]	6.6 [0.4, 25.4]	17.1 [5.7, 40.9]	5.3 [0.0, 25.6]	0.013
Never smoker	122 (36.0%)	83 (38.8%)	24 (30.4%)	15 (32.6%)	0.058
Ex-smoker	149 (44.0%)	83 (38.8%)	39 (49.4%)	27 (58.7%)	
Smoker	68 (20.1%)	48 (22.4%)	16 (20.3%)	4 (8.7%)	

Values are given as arithmetic mean \pm standard deviation for continuous variables, unless indicated otherwise. Categorical data are given as counts (percentage). P-values from one-way ANOVA, Kruskal-Wallis Test or χ 2-Test, where appropriate. BMI = Body Mass Index; HbA1c = Glycated Haemoglobin; LDL = Low Density Lipoprotein

https://doi.org/10.1371/journal.pdig.0000429.t001

Deep learning framework for liver shape analysis

In the independent testing dataset, the deep learning liver segmentation model showed a high performance with a mean Dice score of 97.0 \pm 0.01, an average surface distance of a 0.82 \pm 0.19 mm and a Pearson's correlation coefficient of 0.998 (p<0.001) (Fig 3) compared to the manual labels generated by the expert radiologist.

Visual inspection of all participants revealed no systemic segmentation errors or major failures. From all segmentation masks, the entire panel of 14 radiomic shape features were successfully extracted. An overview and a correlation matrix of the features is provided in S1 Fig.



Expert Manual Segmentation (Volume/ml)

Fig 3. Pearson's correlation between expert manual vs. automatic deep learning segmentations. Pearson's correlation between expert manual and automatic deep learning segmentations of the liver indicating a high model performance with a correlation coefficient of r = 1.0 (p<0.001) in the independent testing dataset.

https://doi.org/10.1371/journal.pdig.0000429.g003

Association between liver shape features and impaired glucose metabolism

Clustering approach: First, we investigated the association between all 14 shape features and impaired glucose metabolism using a cluster-based approach. The 3 identified clusters comprised n = 115 (cluster 1), n = 138 (cluster 2) and n = 86 (cluster 3) participants, respectively. Participants in the high-risk cluster (cluster 3) were more likely men, presented with hypertension and hepatic steatosis compared to cluster 1 and 2, (all p<0.001).

On a univariable level, there was an independent association between cluster 2 and cluster 3 and impaired glucose metabolism (OR 1.84 [95% CI 1.08–3.19] and OR 3.86 [95% CI 2.17–7.00]; p < 0.001) compared to cluster 1. The association remained robust after adjustment for age and sex Fig 4. After further adjustment for the complete panels of cardiometabolic risk factors, only the OR for cluster 3 remained significant (OR 2.44 [95% CI 1.12–5.38]; p = 0.026) whereas the signal for cluster 2 was attenuated (OR 1.67 [95% CI 0.88–3.20]; p = 0.121). A summary is provided in Fig 4.

Individual approach: Next, we investigated the association between individual shape features and impaired glucose metabolism. A summary and correlation matrix of the 14 extracted features are given in S1 Fig and S1 Table. Surface Area, Voxel Volume, Least-, Minor-, Major Axis Length, Surface Volume Ratio, Maximum 2D Diameter Row and Mesh Volume were significantly associated with impaired glucose metabolism while no significant association was found for Elongation, Sphericity, Flatness, Maximum2D Diameter Column, Maximum 2D Diameter Slice and Maximum 3D Diameter (Fig 5). Applying the feature reduction approach described above (OR \geq 1.5 and correlation with liver volume \leq 0.5), liver volume remained as the only feature independently associated with impaired glucose metabolism on a univariable level (OR 2.04 [95% CI 1.63–2.50]; p = 0.001). This association remained significant after adjustment for age and sex and the complete panel of cardiometabolic risk factors (OR 1.97 [1.38–2.85]; p<0.001) (Fig 5).



Fig 4. Association between hepatic shape feature clusters and impaired glucose metabolism. Association between hepatic shape features clusters (A = intermediate risk cluster; B = high-risk cluster) and impaired glucose metabolism (prediabetes or diabetes) from ordinal logistic regression with proportional odds assumption. Model 1 = univariate; Model 2 = adjusted for age, sex; Model 3 = adjusted age, sex and hepatic steatosis; Model 4 = adjusted for age, sex, BMI, alcohol consumption, hypertension, smoking (never, ex, current), total cholesterol and hepatic steatosis. Boxes indicate odds ratios; lines 95% confidence intervals. BMI = Body Mass Index.

https://doi.org/10.1371/journal.pdig.0000429.g004



Fig 5. Association between individual hepatic shape features with impaired glucose metabolism and association between liver volume and impaired glucose metabolism. A) Association between individual hepatic shape features and impaired glucose metabolism (prediabetes or diabetes) from ordinal logistic regression with proportional odds assumption. Figure depicts univariate odds ratios with 95% confidence intervals for hepatic shape features association with impaired glucose metabolism. B) Association between liver volume and impaired glucose metabolism (prediabetes or diabetes) from ordinal logistic regression with proportional odds assumption. Model 1 = univariate; Model 2 = adjusted for age, sex; Model 3 = adjusted age, sex and hepatic steatosis; Model 4 = adjusted for age, sex, BMI, alcohol consumption, hypertension, smoking (never, ex, current), total cholesterol and hepatic steatosis. Boxes indicate odds ratios; lines 95% confidence intervals. BMI = Body Mass Index.

https://doi.org/10.1371/journal.pdig.0000429.g005

Discussion

In this study, we developed a fully automatic deep learning framework to investigate the association between radiomic shape features of the liver and impaired glucose metabolism. Our results demonstrate that MRI-derived radiomic shape features of the liver are independently associated with impaired glucose metabolism after adjustment for traditional cardiometabolic risk factors and hepatic steatosis in a cohort of individuals with prediabetes, type 2 diabetes and normal controls.

Currently, several accurate and cheap tests exist for the diagnosis and monitoring of impaired glucose metabolism. Nevertheless, the number of individuals with undiagnosed diabetes is high with an estimate of 44.7% in 2021 [9]. This number is of great socioeconomic concern given the costs associated with the treatment of diabetes and its complications which are estimated by Bommer et al. to accumulate to 2.1 trillion dollars in 2030 [5]. Hence, screening and early detection are of great importance to identify individuals in an asymptomatic stage to reduce the need for intensified treatment and management of potentially irreversible complications since dietary changes and physical activity have proven to be sufficient preventive measures in the early disease stage [23]. Therefore, novel approaches to identify individuals at risk of type 2 diabetes are desirable to reduce morbidity and mortality.

In this context, opportunistic risk assessment based on routinely acquired but currently unused imaging data may provide a valuable solution. While vast amounts of imaging data are acquired every day, only a small fraction is used to rule out or confirm a suspected clinical finding. Quantification of prognostic imaging information, such as coronary artery calcium or liver fat is only done for specific indications requiring dedicated equipment and expertise [24] [25]. With recent advances in deep learning, automatic quantification of such information has become feasible at high speed and low additional cost. For example, Zeleznik et al. proposed a deep learning model for automatic quantification of coronary artery calcium and

demonstrated a robust performance in more than 20.000 individuals [26]. Graffy et al. presented a method for population-based assessment of hepatic steatosis on CT imaging in more than 9000 subjects [27]. In line with these studies, our results demonstrated the potential for opportunistic screening of impaired glucose metabolism using a standard MRI of the liver in a cohort of 339 individuals. As MRI of the liver is frequently acquired in clinical routine, our framework could add an additional layer of information to the standard radiology report with a personalized risk estimate for impaired glucose metabolism. By implementing the model in the Picture Archiving and Communication System, this could be done with minimal disruption of current workflows and provide valuable prognostic information to the patient and treating physician that currently goes unnoticed. If an increased risk is detected, the patient could be notified to initiate risk factor assessment and lifestyle interventions.

The literature on opportunistic risk assessment of impaired glucose metabolism using medical imaging data is limited. We identified only two studies investigating the value of CT-based radiomic features of the liver and the pancreas to estimate the risk of diabetes [28,29]. In contrast to the current study, the identified features/signatures comprised various features, which are not intuitively understood and generalizability as well as stability to other populations is questionable. A more simple and known measure like liver volume identified in our analysis is easier to grasp for the referring physician and the patient alike, which might increase acceptance, as the black box characteristic of many deep learning approaches is a well-known drawback and limitation for clinical implementation [30,31].

Several limitations of this study need to be considered. First, our study population comprised predominantly Caucasian subjects without known cardiovascular disease. Whether our results generalize to other races/ethnicities and individuals with prevalent cardiovascular disease needs to be tested in more diverse populations. Further, the deep learning framework was not externally validated. Thorough testing and potential retraining/transfer learning might be necessary to ensure reliable performance in other populations. Moreover, we only explored the association between radiomic shape features of the liver. Whether higher-level features carry further additive value was beyond the scope of this study. Finally, although our approach identified individuals at risk of impaired glucose metabolisms it remains unknown whether this would change clinical workflows for disease prevention and patient management.

In conclusion, the proposed deep learning framework allows for identifying individuals at increased risk of impaired glucose metabolism on routinely acquired MRI of the liver independent of traditional cardiometabolic risk factors and hepatic steatosis. This approach could be used for opportunistic screening in daily care to prompt risk factor assessment and initiate preventive measures to reduce morbidity and mortality.

Supporting information

S1 Table. Radiomic shape parameters of our study population. (DOCX)

S2 Table. Baseline demographics and cardiovascular risk factors for the clusters. (DOCX)

S1 Fig. Pearson's correlation coefficient between the extracted radiomic shape features of the liver. 1 indicates a perfect correlation. (TIF)

Acknowledgments

We thank all participants for their long-term commitment to the KORA study, the staff for data collection and research data management and the members of the KORA Study Group (https://www.helmholtz-munich.de/en/epi/cohort/kora) who are responsible for the design and conduct of the study.

Author Contributions

Conceptualization: Lea J. Michel, Marco Reisert, Alexander Rau, Jakob Weiss.

Data curation: Lea J. Michel, Susanne Rospleszcz.

Formal analysis: Susanne Rospleszcz, Marco Reisert.

Software: Marco Reisert.

Supervision: Susanne Rospleszcz, Alexander Rau, Johanna Nattenmueller, Fabian Bamberg, Jakob Weiss.

Validation: Lea J. Michel, Susanne Rospleszcz, Jakob Weiss.

Writing - original draft: Lea J. Michel, Jakob Weiss.

Writing – review & editing: Susanne Rospleszcz, Marco Reisert, Alexander Rau, Johanna Nattenmueller, Wolfgang Rathmann, Christopher. L. Schlett, Annette Peters, Fabian Bamberg, Jakob Weiss.

References

- Chatterjee S, Khunti K, Davies MJ. Type 2 diabetes. The Lancet [Internet]. 2017 Jun [cited 2022 Jun 18]; 389(10085):2239–51. Available from: https://linkinghub.elsevier.com/retrieve/pii/ S0140673617300582
- Shaw JE, Sicree RA, Zimmet PZ. Global estimates of the prevalence of diabetes for 2010 and 2030. Diabetes Res Clin Pract [Internet]. 2010 Jan [cited 2022 Sep 30]; 87(1):4–14. Available from: https:// linkinghub.elsevier.com/retrieve/pii/S016882270900432X https://doi.org/10.1016/j.diabres.2009.10. 007 PMID: 19896746
- Cho NH, Shaw JE, Karuranga S, Huang Y, da Rocha Fernandes JD, Ohlrogge AW, et al. IDF Diabetes Atlas: Global estimates of diabetes prevalence for 2017 and projections for 2045. Diabetes Res Clin Pract [Internet]. 2018 Apr [cited 2022 Sep 30]; 138:271–81. Available from: https://linkinghub.elsevier. com/retrieve/pii/S0168822718302031 https://doi.org/10.1016/j.diabres.2018.02.023 PMID: 29496507
- 4. Sun H, Saeedi P, Karuranga S, Pinkepank M, Ogurtsova K, Duncan BB, et al. IDF Diabetes Atlas: Global, regional and country-level diabetes prevalence estimates for 2021 and projections for 2045. Diabetes Res Clin Pract [Internet]. 2022 Jan [cited 2022 Jun 18]; 183:109119. Available from: https:// linkinghub.elsevier.com/retrieve/pii/S0168822721004782 https://doi.org/10.1016/j.diabres.2021. 109119 PMID: 34879977
- Bommer C, Sagalova V, Heesemann E, Manne-Goehler J, Atun R, Bärnighausen T, et al. Global Economic Burden of Diabetes in Adults: Projections From 2015 to 2030. Diabetes Care [Internet]. 2018 May 1 [cited 2022 Sep 30]; 41(5):963–70. Available from: https://diabetesjournals.org/care/article/41/5/ 963/36522/Global-Economic-Burden-of-Diabetes-in-Adults https://doi.org/10.2337/dc17-1962 PMID: 29475843
- Jacobs E, Hoyer A, Brinks R, Icks A, Kuß O, Rathmann W. Healthcare costs of Type 2 diabetes in Germany. Diabet Med [Internet]. 2017 Jun [cited 2022 Jun 18]; 34(6):855–61. Available from: https://doi.org/10.1111/dme.13336 PMID: 28199029
- 7. Economic Costs of Diabetes in the U.S. in 2017. 2018; 41:12.
- National Diabetes Statistics Report 2020. Estimates of diabetes and its burden in the United States. 2020;32.
- Ogurtsova K, Guariguata L, Barengo NC, Ruiz PLD, Sacre JW, Karuranga S, et al. IDF diabetes Atlas: Global estimates of undiagnosed diabetes in adults for 2021. Diabetes Res Clin Pract [Internet]. 2022 Jan [cited 2022 Jun 18]; 183:109118. Available from: https://linkinghub.elsevier.com/retrieve/pii/ S0168822721004770 https://doi.org/10.1016/j.diabres.2021.109118 PMID: 34883189

- Hutchinson MS, Joakimsen RM, Njølstad I, Schirmer H, Figenschau Y, Svartberg J, et al. Effects of Age and Sex on Estimated Diabetes Prevalence Using Different Diagnostic Criteria: The Tromsø OGTT Study. Int J Endocrinol [Internet]. 2013 [cited 2022 Jun 19]; 2013:1–9. Available from: http://www. hindawi.com/journals/ije/2013/613475/
- Hartwig S, Kluttig A, Tiller D, Fricke J, Müller G, Schipf S, et al. Anthropometric markers and their association with incident type 2 diabetes mellitus: which marker is best for prediction? Pooled analysis of four German population-based cohort studies and comparison with a nationwide cohort study. BMJ Open [Internet]. 2016 Jan [cited 2022 Jun 19]; 6(1):e009266. Available from: https://bmjopen.bmj.com/lookup/doi/10.1136/bmjopen-2015-009266 PMID: 26792214
- Nakagami T, Qiao Q, Tuomilehto J, Balkau B, Carstensen B, Tajima N, et al. The fasting plasma glucose cut-point predicting a diabetic 2-h OGTT glucose level depends on the phenotype. Diabetes Res Clin Pract [Internet]. 2002 Jan [cited 2022 Jun 19]; 55(1):35–43. Available from: https://linkinghub. elsevier.com/retrieve/pii/S0168822701002704 https://doi.org/10.1016/s0168-8227(01)00270-4 PMID: 11755477
- Guo F, Moellering DR, Garvey WT. Use of HbA1c for Diagnoses of Diabetes and Prediabetes: Comparison with Diagnoses Based on Fasting and 2-Hr Glucose Values and Effects of Gender, Race, and Age. Metab Syndr Relat Disord [Internet]. 2014 Jun [cited 2022 Jun 19]; 12(5):258–68. Available from: http:// www.liebertpub.com/doi/10.1089/met.2013.0128 PMID: 24512556
- Kengne AP, Beulens JW, Peelen LM, Moons KG, van der Schouw YT, Schulze MB, et al. Non-invasive risk scores for prediction of type 2 diabetes (EPIC-InterAct): a validation of existing models. Lancet Diabetes Endocrinol [Internet]. 2014 Jan [cited 2022 Jun 19]; 2(1):19–29. Available from: https://linkinghub. elsevier.com/retrieve/pii/S2213858713701037 https://doi.org/10.1016/S2213-8587(13)70103-7 PMID: 24622666
- Herder C, Kowall B, Tabak AG, Rathmann W. The potential of novel biomarkers to improve risk prediction of type 2 diabetes. Diabetologia [Internet]. 2014 Jan [cited 2022 Jun 19]; 57(1):16–29. Available from: http://link.springer.com/10.1007/s00125-013-3061-3 PMID: 24078135
- Hippisley-Cox J, Coupland C, Robson J, Sheikh A, Brindle P. Predicting risk of type 2 diabetes in England and Wales: prospective derivation and validation of QDScore. BMJ [Internet]. 2009 Mar 17 [cited 2022 Sep 30];338(mar17 2):b880–b880. Available from: https://www.bmj.com/lookup/doi/10. 1136/bmj.b880 PMID: 19297312
- Rosella LC, Manuel DG, Burchill C, Stukel TA, for the PHIAT-DM team. A population-based risk algorithm for the development of diabetes: development and validation of the Diabetes Population Risk Tool (DPoRT). J Epidemiol Community Health [Internet]. 2011 Jul 1 [cited 2022 Sep 30]; 65(7):613–20. Available from: https://jech.bmj.com/lookup/doi/10.1136/jech.2009.102244 PMID: 20515896
- Holle R, Happich M, Löwel H, Wichmann H. KORA—A Research Platform for Population Based Health Research. Gesundheitswesen [Internet]. 2005 Aug [cited 2022 Jun 12];67(S 01):19–25. Available from: http://www.thieme-connect.de/DOI/DOI?10.1055/s-2005-858235 PMID: 16032513
- Bamberg F, Hetterich H, Rospleszcz S, Lorbeer R, Auweter SD, Schlett CL, et al. Subclinical Disease Burden as Assessed by Whole-Body MRI in Subjects With Prediabetes, Subjects With Diabetes, and Normal Control Subjects From the General Population: The KORA-MRI Study. Diabetes [Internet]. 2017 Jan 1 [cited 2022 Jun 12]; 66(1):158–69. Available from: https://diabetesjournals.org/diabetes/ article/66/1/158/40108/Subclinical-Disease-Burden-as-Assessed-by-Whole https://doi.org/10.2337/ db16-0630 PMID: 27999110
- Reisert M, Russe M, Elsheikh S, Kellner E, Skibbe H. Deep Neural Patchworks: Coping with Large Segmentation Tasks [Internet]. arXiv; 2022 [cited 2022 Sep 30]. Available from: http://arxiv.org/abs/2206. 03210
- Çiçek Ö, Abdulkadir A, Lienkamp SS, Brox T, Ronneberger O. 3D U-Net: Learning Dense Volumetric Segmentation from Sparse Annotation. In: Ourselin S, Joskowicz L, Sabuncu MR, Unal G, Wells W, editors. Medical Image Computing and Computer-Assisted Intervention–MICCAI 2016 [Internet]. Cham: Springer International Publishing; 2016 [cited 2022 Sep 29]. p. 424–32. (Lecture Notes in Computer Science; vol. 9901). Available from: https://link.springer.com/10.1007/978-3-319-46723-8_49
- World Health Organization, International Diabetes Federation. Definition and diagnosis of diabetes mellitus and intermediate hyperglycaemia: report of a WHO/IDF consultation. 2006 [cited 2022 Jun 12]; Available from: https://apps.who.int/iris/handle/10665/43588
- 23. Uusitupa M, Khan TA, Viguiliouk E, Kahleova H, Rivellese AA, Hermansen K, et al. Prevention of Type 2 Diabetes by Lifestyle Changes: A Systematic Review and Meta-Analysis. Nutrients [Internet]. 2019 Nov 1 [cited 2022 Sep 30]; 11(11):2611. Available from: https://www.mdpi.com/2072-6643/11/11/2611 https://doi.org/10.3390/nu11112611 PMID: 31683759
- Kantor B, Nagel E, Schoenhagen P, Barkhausen J, Gerber TC. Coronary Computed Tomography and Magnetic Resonance Imaging. Curr Probl Cardiol [Internet]. 2009 Apr [cited 2022 Nov 10]; 34(4):145–

217. Available from: https://linkinghub.elsevier.com/retrieve/pii/S0146280608001849 https://doi.org/10. 1016/j.cpcardiol.2008.12.002 PMID: 19269527

- Reeder SB, Cruite I, Hamilton G, Sirlin CB. Quantitative assessment of liver fat with magnetic resonance imaging and spectroscopy. J Magn Reson Imaging [Internet]. 2011 Oct [cited 2022 Nov 10]; 34 (4):729–49. Available from: https://onlinelibrary.wiley.com/doi/ https://doi.org/10.1002/jmri.22775 PMID: 22025886
- Zeleznik R, Foldyna B, Eslami P, Weiss J, Alexander I, Taron J, et al. Deep convolutional neural networks to predict cardiovascular risk from computed tomography. Nat Commun [Internet]. 2021 Dec [cited 2022 Sep 30]; 12(1):715. Available from: http://www.nature.com/articles/s41467-021-20966-2 https://doi.org/10.1038/s41467-021-20966-2 PMID: 33514711
- Graffy PM, Sandfort V, Summers RM, Pickhardt PJ. Automated Liver Fat Quantification at Nonenhanced Abdominal CT for Population-based Steatosis Assessment. Radiology [Internet]. 2019 Nov [cited 2022 Sep 30]; 293(2):334–42. Available from: <u>http://pubs.rsna.org/doi/10.1148/radiol.</u> 2019190512 PMID: 31526254
- Lu CQ, Wang YC, Meng XP, Zhao HT, Zeng CH, Xu W, et al. Diabetes risk assessment with imaging: a radiomics study of abdominal CT. Eur Radiol [Internet]. 2019 May [cited 2022 May 14]; 29(5):2233–42. Available from: http://link.springer.com/10.1007/s00330-018-5865-5 PMID: 30523453
- Jang S, Kim JH, Choi SY, Park SJ, Han JK. Application of computerized 3D-CT texture analysis of pancreas for the assessment of patients with diabetes. Hu C, editor. PLOS ONE [Internet]. 2020 Jan 13 [cited 2022 Jul 31]; 15(1):e0227492. Available from: https://doi.org/10.1371/journal.pone.0227492 PMID: 31929591
- Wichmann JL, Willemink MJ, De Cecco CN. Artificial Intelligence and Machine Learning in Radiology: Current State and Considerations for Routine Clinical Implementation. Invest Radiol [Internet]. 2020 Sep [cited 2022 Sep 30]; 55(9):619–27. Available from: <u>https://journals.lww.com/10.1097/RLI</u>. 00000000000673 PMID: 32776769
- Neri E, Coppola F, Miele V, Bibbolino C, Grassi R. Artificial intelligence: Who is responsible for the diagnosis? Radiol Med (Torino) [Internet]. 2020 Jun [cited 2022 Sep 30]; 125(6):517–21. Available from: http://link.springer.com/10.1007/s11547-020-01135-9 PMID: 32006241